Contents lists available at ScienceDirect

Automation in Construction

journal homepage: www.elsevier.com/locate/autcon

Review Unmanned aerial vehicle-based as-built surveys of buildings

Tianzhi He^a, Kaiwen Chen^b, Farrokh Jazizadeh^{c,*}, Georg Reichard^d

^a Via Department of Civil and Environmental Engineering, Virginia Tech, 315 Patton Hall, 750 Drillfield, Blacksburg, VA 24061, United States of America
 ^b Department of Civil, Construction, Environmental Engineering, University of Alabama, 3043 H.M. Comer, 245 7th Avenue, Tuscaloosa, AL 35487, United States of

^c Via Department of Civil and Environmental Engineering, Virginia Tech, 200 Patton Hall, 750 Drillfield, Blacksburg, VA 24061, United States of America ^d Department of Building Construction, Virginia Polytechnic Institute and State University, 1345 Perry St, Blacksburg, VA 24061, United States of America

ARTICLE INFO

America

Keywords: Unmanned Aerial Vehicles (UAVs) As-built building survey Photogrammetry Building inspection 3D reconstruction Building point cloud Digital twins

ABSTRACT

Camera-equipped Unmanned Aerial Vehicles (UAVs) are effective tools for as-built building surveys. Through a systematic literature review, we synthesized and categorized factors affecting the quality of UAV-based reconstructed scenes and their associated performance evaluation metrics (e.g., accuracy and time efficiency). Their interrelationships were analyzed by Social Network Analysis (SNA) to identify critical factors and their impacts. We further quantitatively evaluated these factors and metrics through controlled experiments with a camera-equipped UAV and a Terrestrial Laser Scanner for an institutional building. Various flight paths, photo overlaps, and distances to the building were tested to evaluate their impact on dimensional accuracy, time efficiency, and point cloud density. We demonstrated the trade-offs between the influential factors to provide insights into parameter selection. Additionally, a data requirements schema for UAV-based as-built 3D scene reconstruction was established toward standardization of data processing practices. This study could serve as a foundation for future research and applications of UAV-based photogrammetry for building surveys.

1. Introduction

Unmanned Aerial Vehicles (UAVs) have become increasingly popular in the Architecture, Engineering, and Construction (AEC) industry, with their utilization on project job sites nearly doubling from 20.7% in 2015 to 43% in 2020 [35,36]. UAVs have various applications throughout projects' life cycles, including land surveying, logistics tracking, on-site monitoring, and as-built surveys during the construction phase, as well as applications for maintenance, operation, and ultimately, demolition during the post-construction phase [49]. Among these applications, as-built building surveys are widely used for documenting building information, developing digital twins, and assessing and planning maintenance needs. Compared with conventional manual as-built surveys, three-dimensional (3D) scene reconstruction provides a more efficient and cost-effective solution with detailed visual information and measurements [42]. Although Terrestrial Laser Scanning (TLS) technology could largely reduce the surveying workload and generate a 3D building model with high efficiency and accuracy [70], the high cost of professional equipment limits its wider adoption. In contrast, UAVbased photogrammetry through image processing is an alternative for automated 3D as-built surveys with the benefits of time- and costefficiency. Compared with terrestrial measurement methods, a lowcost camera-equipped UAV system enables the collection of large amounts of building images from multiple views within a short time. These images can later be used for the reconstruction and survey of a 3D building model with access to detailed visual information and measurements [8,20,88].

With UAV-captured images and their camera position information, 3D reconstructed scenes or 2D orthophoto mosaics can be acquired through processing 2D image sets with triangulation and photogrammetric techniques such as Structure-from-Motion (SfM) [67]. To enhance the efficiency (e.g., time and cost) and performance (e.g., dimensional accuracy and point density) of the 3D reconstruction processes, previous research efforts have studied the impact of different data collection and processing features. These include flight design features, such as flight paths, flight timing, altitude, distance to the target, the total number of photos, and photo overlap percentages, as well as ground control points (GCPs) and rocessing software tools. They have shown wide ranges of variations, with the photo overlap percentages ranging from 35% to 95%, and the distance between UAV and building surface from 1 to 25 m for different building inspection applications [73]. However, existing studies have generally discussed overall

* Corresponding author. *E-mail addresses:* tianzhi@vt.edu (T. He), kaiwen.chen@ua.edu (K. Chen), jazizade@vt.edu (F. Jazizadeh), reichard@vt.edu (G. Reichard).

https://doi.org/10.1016/j.autcon.2024.105323

Received 25 August 2023; Received in revised form 12 December 2023; Accepted 6 February 2024 0926-5805/© 2024 Elsevier B.V. All rights reserved.







factors or partially evaluated the influence of specific factors on the performance of UAV photogrammetry. There has been a lack of systematic synthesis of the reported factors and their impacts on performance across different studies. Therefore, this paper uses a systematic literature review and Social Network Analysis (SNA) to synthesize the influential factors, performance evaluation metrics, and their interrelationships for UAV-based as-built surveys for buildings. The experimental quantification of how these identified factors influence the quality of the reconstructed scenes was also carried out to reveal their quantitative relationship and build a foundation for the application of automated UAV-based building surveys. As such, with the aim of investigating these factors interrelationships, this study has sought to answer the following questions:

- What are the factors that impact the performance (i.e., accuracyefficiency trade-off) of UAV-based photogrammetry for the reconstruction of 3D building models?
- What is the quantified impact of these factors on the quality of reconstructed models?

To answer these questions, we first identified the important factors and indicators for performance evaluation through a systematic literature review (SLR) and synthesis. For selected factors that were identified as important in previous literature, controlled experiments were conducted to quantify their accuracy-efficiency trade-off. Therefore, this study provides a comprehensive review of the influential factors and their quantified impact on the as-built survey process and outcome with a case study. The paper is organized into several sections: Section 2 provides an overall review of previous studies related to UAV applications in construction surveying and 3D reconstruction. Section 3 introduces the methodology of this study, which includes the SLR and the controlled experiments on a real-world building. Section 4 presents the results of the SLR, which identifies the critical influential factors, as well as metrics for performance evaluation. Section 5 discusses the case study including the selected influential factors and performance indicators, the controlled experiments, and the descriptive statistical analysis. Finally, Section 6 summarizes the key findings and limitations, and describes the future directions.

2. Background

2.1. Photogrammetry in the AEC industry

A photogrammetric survey entails mathematical modeling through central projection imaging techniques to locate object points in three dimensions and thus form 3D reconstructed scenes based on 2D images [54]. In the AEC applications, photogrammetry is typically referred to as close-range photogrammetry, where the camera is positioned within a range of 1 to 300 m from the target object [14]. The dimensions of the target object usually range from 0.5 to 200 m with an accuracy ranging from 0.1 mm to 1 cm [54]. In the AEC industry, photogrammetric surveys have been broadly applied in various stages of lifecycle management. For example, during the construction phase, photogrammetry can be utilized to generate consecutive as-built point clouds to monitor the progress of a construction project. By measuring and comparing the dimensions from the generated point clouds and the designed 4D building information model (4D-BIM), construction progress monitoring could be conducted with higher accuracy [80,89]. Photogrammetry has also been adopted in operational stages to model building facades for surveys and inspections [66,95].

There are a variety of factors that can influence the performance of photogrammetric surveys. Terrestrial photogrammetry, a subset of general photogrammetry, utilizes ground-based cameras and imaging techniques to capture data and measure the three-dimensional information of objects and environments [53]. It typically relies on specific sensing equipment and techniques tailored for ground-based data

collection. Dai et al. [13,15] systematically reviewed the literature related to terrestrial photogrammetric modeling to identify influential factors, including camera-object distance, the number of photos, intersection angle, angle of incidence, and camera-related features (i.e., camera model, resolution, focal length, camera lens). While terrestrial photogrammetry has demonstrated significant potential for building surveys, its inherent limitations, notably the constrained field of view, hinder the surveying of geometric objects that are inaccessible, such as high-elevation building facades and roofs [75].

2.2. UAV applications in the AEC industry

With advances in UAV technologies, the application of cost-effective UAV camera systems could augment the data acquisition processes with improved accessibility and efficiency. This, in particular, enhances the capability for surveying and monitoring building structures at higher elevations expanding potential applications. In addition, UAVs can also be applied in indoor environments for applications such as monitoring construction sites, surveying, and information collection [38,48,57]. As revealed in previous studies, UAVs have found extensive applications across various stages of a building's life cycle. In the construction phase, UAVs can be used for detecting changes and deviations from planned structures [33]. Furthermore, in the post-construction phase, UAVs play a vital role in inspecting [43,81,85], measuring [31,81], and documenting the as-built model of construction projects [2]. In addition, recent studies have explored the utilization of multiple UAVs for the collection and analysis of large-scale data about multiple buildings, highlighting their potential for enhancing efficiency and precision in large-scale urban planning and development projects [10,28,50,97]. Several studies have reviewed the developments and applications of UAVs in the AEC industry. As shown in Table 1, we have compiled and summarized the main findings of these review studies. Most of these review papers introduced the application areas in the construction industry [5,8,16,59,102]. Some of them summarized the technologies, benefits, and challenges of applying UAVs in different fields [8,21,72,99,102]. Nevertheless, limited studies have compiled and investigated the influential factors in the UAV-based as-built 3D reconstruction of buildings [71,73].

Among all categories of UAV applications in the AEC industry, the focus of this study lies in the use of UAVs for as-built building surveys (i. e., 3D reconstructed models of buildings). Ham et al. [30] have compiled the related literature in this direction to summarize the workflow of UAV photogrammetry in building surveys as three major steps: (1) UAVbased data collection, (2) data processing to obtain actionable information (e.g., point clouds and reconstructed scenes or models), and (3) visualization, evaluation, and communication with practitioners in the project. These steps have been used in our study for the categorization of influential factors and evaluation metrics. Although the implementation of UAVs has greatly improved the application of photogrammetry for 3D building surveys and researchers have thoroughly reviewed and categorized these applications as reflected in Table 1, few studies have explored the influential factors that affect the outcome of the process in both qualitative and quantitative ways. In practical applications, while numerous case studies have demonstrated successful high-quality 3D scene reconstructions, there remains a gap in our collective understanding of how variations in individual influential factors affect the quality of building surveys across different levels of detail [15,71]. Despite some factors that are more or less similar in terrestrial photogrammetry (e.g., the distance between the camera and targets or the number of photos), UAV-based photogrammetric surveys have their unique features that need to be further investigated (e.g., flight path and flight height).

3. Research methodology

Our methodology was divided into two components: (1) a systematic

Key review papers of UAV applications in the AEC industry.

References	Year	Main Findings
[8]	2014	 UAV hardware, software, control methodologies, and the latest related technologies. Opportunities and challenges of applying UAV in 1) seismic risk assessment, 2) transportation, 3) disaster response, 4) construction management, 5) surveying and mapping, and 6) flood monitoring and assessment.
[30]	2016	A review of UAV-driven research on automating construc- tion monitoring and civil infrastructure condition assessment
[19]	2017	 A value chain with stakeholders involved in UAV applications to outline the major trends of UAV usage in the industry. Suggested four applications and gaps in the use of UAVs in 1) the design phase, 2) the construction phase, and 3) the
[57]	2017	 The potential applications for UAVs in indoor construction sites, their benefits, and challenges.
[86]	2017	 Summarized four categories of UAV use in the construction industry: 1) photography/videography, 2) surveying, 3) inspections, and 4) safety/security monitoring.
[60]	2017	 Characterized 9 applications and 10 issues reported in previous studies of applying UAVs in the construction industry
[73]	2018	 A historical timeline of UAV technology developments. A standard procedure for operating a UAV for energy audit missions (documenting building performance, visualizing heat transfer using infrared imaging, and creating digital models).
[16]	2018	 Identified five Applications of UAV in the construction industry: 1) Project progress control, 2) Damage assessment, 3) Surveying, 4) Safety monitoring, 5) 3D modeling
[102]	2018	 Applications of UAVs in the industry: 1) building inspection, 2) damage assessment, 3) site surveying, 4) safety inspection, 5) progress monitoring, 6) building maintenance, and 7) other construction applications. Benefits, types, onboard sensors, and control styles of UAVs in construction applications.
[5]	2019	 Categorization of UAV applications in the AEC domain: 1) infrastructure and structural inspection, 2) transportation, 3) cultural heritage monitoring, 4) city and urban planning, 5) progress monitoring, 6) post-disaster, 7) construction safety
[99]	2020	 Eight potential UAV application areas and three major challenges were identified. The challenges include 1) legal and regulatory requirements, 2) features and abilities of UAVs, and 3) qualities and capabilities of related software.
[21]	2020	• Four areas of applications of immersive technologies and two areas of applications of UAVs in the construction industry. For UAVs, the areas include 1) automated surveying, information management, and visualization, 2) construction inspection, monitoring, and safety management.
[100]	2021	 Two key components of applications of UAV aerial images for three-dimensional reconstruction: 1) image-based three-dimensional reconstruction technology, 2) UAV path planning technology.
[94]	2021	 Five main UAV application fields: 1) Mapping and 3D modeling, 2) Construction monitoring, 3) Structural damage detection, 4) Energy efficiency prospection, 5) Urban remote sensing.
[72]	2022	• Seven application areas of integrating UAV with Digital Twins: 1) progress monitoring, 2) historic building conservation, 3) information management, 4) construction safety, 5) construction education, 6) structural and infrastructure inspection, 7) transportation. Four technology trends: 1) automated progress monitoring, 2) automated UAV inspection planning, 3) real-time video streaming, 4) parametric model development of historic buildings
[65]	2023	Five main dimensions of UAV application in the AEC

industry: 1) inspections and mapping, 2) data processing and management, 3) safety and health management, 4)

Table 1 (con	tinued)	
References	Year	Main Findings
[59]	2023	 challenges and risks of drone use, 5) training aid for site personnel and students Market review and comparison among different UAV types Future directions of research: education, total quality management, safety risk related to drone use, human factors and their impact, regulatory interventions. Two categories of UAV applications in construction management: 1) UAV uses including algorithms, applications, operations, framework, and training, 2) Construction uses including inspection, surveying, safety, and monitoring.

literature review and synthesis and (2) an experimental case study. To assess the underlying factors affecting the quality of the reconstructed scenes, we compiled the related articles and summarized the reported influential factors. These articles present field studies with quantified assessments of 3D scenes. In the next step, we analyzed these factors to develop a categorization for them. To explore the reflected relationships between different factors, we used the analytical method of Social Network Analysis (SNA) to measure the centrality degrees of the factors that appeared in the compiled articles. This method could be used to identify the co-occurrence frequency of different factors in previous studies and visualize their interrelationships [55].

After identifying the characteristics of influencing factors and their interrelationships, a case study with controlled experiments on a university building was conducted to quantitatively evaluate the relationship between the identified important factors. The impact of three influencing factors (the flight path pattern, the distance between the UAV and the target, and the photo overlap percentage) was investigated by using three performance evaluation metrics including dimensional accuracy, point cloud density, and time efficiency. To this end, six mesh models of the reconstructed scenes with different settings were developed. Additionally, a terrestrial laser scanner-based (TLS-based) point cloud was also developed as the benchmark for performance evaluation. Benchmarking against the point clouds from laser scanners is a commonly used approach given the high accuracy of their resultant point clouds [4,25]. Based on the assessment of these models, we drew conclusions and discussed recommendations. Fig. 1 summarizes the methodology of our study. Details of each step have been presented in their corresponding results subsections.

4. Systematic literature review/synthesis

A Systematic Literature Review (SLR) was conducted to identify, select, and appraise the literature relevant to UAV-based photogrammetry for 3D as-built surveys. Following the Preferred Reporting Items for Systematic Reviews and Meta-Analyses (PRISMA) guidelines [58], the SLR process was conducted as shown in the following steps.

4.1. Literature search strategies

We focused our search on articles with the subject matter at the intersection of UAVs, image-based 3D model reconstruction, and building assets. Considering that there are many alternative terms, the literature databases/libraries and various search keywords were used as follows:

• Literature database/libraries:

- o American Society of Civil Engineers (ASCE) library.
- o Institute of Electrical and Electronics Engineers (IEEE) Xplore.
- o Web of Science (WoS).
- o Engineering Village.



Fig. 1. The methodology of identifying the influential factors in developing 3D reconstructed scenes and their quantified impact.

- Since UAV systems and technologies are relatively new applications in the fields of buildings and construction, we concentrated on studies published in the past decade (2010–2023).
- Search terms
 - o Terms regarding UAV:
 - "UAV" OR "UAS" OR "drone" OR "unmanned aerial vehicle" OR "MAV";
 - o Terms regarding image-based 3D scene reconstruction:
 - "image" OR "photo" AND
 - "model reconstruction" OR "3D reconstruction" OR "photogrammetry";

- o Terms regarding building assets:
 - "building" OR "architecture".

After the compilation, the studies were filtered by inclusion and exclusion criteria listed below:

- Inclusion criteria:
 - o include studies on real-world case studies or experimental practices
 - o include studies that present the complete process of UAV-based 3D scene reconstruction



Fig. 2. Flowchart of the systematic literature review process (PRISMA flow diagram).

T. He et al.

- Exclusion criteria:
 - o Studies that are not in the domain of buildings (e.g., land survey)
 - o Studies that do not include real-world case studies or experimental practices with a complete process of UAV-based 3D scene reconstruction
 - o Studies that are not using UAV-captured RGB images for photogrammetry (e.g., Lidar-based and UGV-based)

Fig. 2 illustrates the detailed procedure of the employed SLR using PRISMA guidelines. A total of 761 articles were identified through the aforementioned search strategy. These articles were then synthesized into 420 articles after removing duplicates and other publication forms (e.g., book chapters and articles in press). Then, the title, abstract, and keywords of 420 articles were further screened using the specified inclusion and exclusion criteria that resulted in the selection of 84 articles.

Table 2

Specification of the studies described in the selected articles related to UAV-based photogrammetric as-built surveys for buildings.

Reference	Year	Building Type	UAV & Camera Model (Resolution)	Mentioned Influential Factors	Mentioned Performance Evaluation Metrics
[34]	2010	Commercial	UAV: MD4-200 Camera: PENTAX Ontio A40 (12MP)	Flight path, GCPs*, Photo overlap, Number of	Number of points, Point cloud
[45]	2011	Commercial	UAV: (1) AscTec Falcon 8 (14MP), (2) swinglet	Flight path, Altitude, Photo Overlap, Number of	Number of points
[26]	2012	School Building	UAV: FKC-1	GCPs, Photo overlap	Scale of mapping
[91]	2013	Cultural	UAV: DJI Phantom	Flight path, GCPs, Processing software	No performance evaluation
[7]	2014	School Building	UAV: 3DR IRIS Quadcopter Camera: GoPro 3 (12MP)	Flight path, GCPs, Photo overlap	Dimensional accuracy
[25]	2014	School Building	UAV: Oben octocopter	Flight path, Altitude, GCPs, Photo overlap, Number of photos Processing software	Point cloud accuracy
[18]	2014	Commercial Building	UAV: DJI Mavic Pro with a camera (12MP)	Flight path, Altitude, Distance, Photo overlap, Number of photos, Processing software	No performance evaluation
[12]	2015	School Building	UAV: AscTech Falcon octocopter Camera: Sony NEX-5 N	Number of photos, Processing software	Point cloud accuracy
[27]	2015	School Building	UAV: Asctec Falcon Camera: Camera with 10MP resolution	Processing software	Façade resolution in pixel
[29]	2015	Cultural Heritage	UAV: MD4–1000 Camera: Olympus PEN <i>E</i> - P2 (12MP)	Flight path, Distance, GCPs, Number of photos, Processing software	Point cloud accuracy
[98]	2015	Residential Building	UAV: DJI Phantom 2	Flight path, Altitude, Number of photos, Processing software	No performance evaluation
[93]	2015	Residential Building	VTOL (vertical take-off and landing) UAV	Number of photos, Processing software	Point cloud density
[4]	2016	Cultural Heritage	UAV: Mikrokopter Camera: Sony ILCE-5100 (24MP)	Flight path, Altitude, GCPs, Processing software	Point cloud accuracy, Number of points
[61]	2017	Cultural Heritage	UAV: (1) Sensefly Albris (2) DJI Phantom 3 Professional (38MP)	Distance, GCPs, Number of photos, Processing software	Point cloud accuracy
[92]	2017	Residential Building	UAV: FlyNovex Camera: Sony Alfa 6000 (24MP)	Flight path, Altitude, GCPs, Photo overlap, Number of photos, Processing software	Dimensional accuracy, Point cloud accuracy
[101]	2018	Stadium (school)	UAV: DJI F550 Camera: GoPro Hero 3 Plus	Flight path, Altitude, Number of photos, Processing software	Point cloud accuracy, Number of points, Time efficiency
[82]	2018	Cultural Heritage	UAV: DJI Phantom 4 (38MP)	Flight path, GCPs, Number of photos, Processing software, Point deviation	Point cloud accuracy
[44]	2019	Residential Building	UAV: DJI Mavic Pro 2 (12MP)	Flight path, Altitude, Distance, Number of photos, Processing software	Point cloud accuracy
[37]	2020	Heritage	UAV: DJI Inspire 1 Camera: Zenmuse X5R (16MP)	Altitude, image shooting rate, Flight speed, Camera angle, Number of photos, Processing software,	Point cloud accuracy
[56]	2020	Cultural	UAV: DJI Matrice 600 Pro	Flight path, GCPs, Altitude, Photo overlap,	Dimensional accuracy, Point
[6]	2020	Gultural	UAV: DJI Phantom Pro 4 with camera (20MP);	GCPs, Altitude, Photo overlap, Number of photos,	Point cloud accuracy
[<mark>53</mark>]	2020	Heritage Cultural	Corby Drone CX012 with camera (0.3MP) UAV: DJI Mavic Pro	Processing software GCPs, Altitude, Distance, Number of photos,	Point cloud accuracy
[51]	2021	Heritage School Building	UAV: DJI Phantom Pro 4 with a camera	Flight path, Distance, Photo overlap, Number of	Dimensional accuracy
[77]	2021	School Building	UAV: DJI Mavic	Flight path, Altitude, Photo overlap, Processing	Dimensional accuracy
[62]	2022	Public Building	UAV type not specified	Flight altitude, Photo overlap, Number of photos,	Point cloud density, Point cloud
[32]	2022	School Building	UAV: DJI phantom 4 PRO with a camera (20MP)	Flight path, Altitude, Photo overlap, Number of photos. Processing software	Point cloud accuracy
[78]	2022	Stadium (public)	UAV: DJI Phantom 3 with a camera (12MP)	GCPs, Processing software	Dimensional accuracy
[87]	2022	School Building	UAV: Autel Evo Nano with a camera (48MP)	Flight path, Altitude, Camera angle, Number of photos, Processing software	Point cloud density
[79]	2023	Residential Building	UAV: DJI Phantom 4 Pro with a camera (20MP)	GCPs, Flight path, Altitude, Distance, Photo overlap, Number of photos, Processing software	Point cloud accuracy
[90]	2023	Cultural Heritage	UAV: DJI Matrice 300 RTK Camera: DJI Zenmuse P1 camera (45MP)	GCPs, Flight path, Altitude, Number of photos, Processing software	Point cloud accuracy

* GCP: Ground Control Points.

** Cloud-cloud distance based on the reference cloud (LiDAR).

After conducting a full-text review of these articles, we extracted a subset of 30 articles that introduced factors used in their UAV-based photogrammetry, and were thus used for the detailed quantitative analysis.

4.2. Literature analysis

The details of the selected 30 articles are listed in Table 2, which summarizes the specifications of each study including the target building type, equipment setup, and reported factors in their experimental design. As shown in this table, previous studies have discussed various influential factors for different scenes. We used these reported factors in developing a qualitative categorization and a quantitative relationship analysis in the following sections.

4.2.1. Categorization of influential factors

The process of using UAVs for photogrammetric building surveys mainly includes the equipment setup (e.g., UAV type, camera resolution, etc.), planning of in-flight data collection, and post-flight processing of collected imagery data [23,73]. Upon review of the articles in Table 2, we clustered the parameters of interest that were discussed in each article. The parameters were respectively categorized into (1) influential factors in the process of 3D scene reconstruction and (2) performance evaluation metrics. Fig. 3 illustrates the mapping of all the parameters of interest. The influential factors can be further grouped into factors related to data collection versus data processing steps. The performance evaluation metrics could be associated with accuracy, time efficiency, and point cloud characteristics. Table 3 shows the range of influential factors reported in the previous studies. These ranges were identified across multiple papers as individual studies have not specified the effective ranges. In the following subsections, further details are presented.

4.2.1.1. Data collection. In the UAV-based data collection processes, multiple factors need to be configured before take-offs, such as the equipment setup and the flight path design. These factors must be set considering the constraints imposed by the UAVs and their cameras, battery life, surveying conditions, and legal regulations [52]. In terms of hardware settings, it can be seen from Table 3, that DJI UAVs are

commonly used multirotor UAVs in photogrammetry. Although, in the literature, there is no clear reported linkage between the resolution or the type of the UAV-attached camera and the quality of the reconstructed scenes, a study in terrestrial photogrammetry indicated that the average error is reduced by 0.83 cm (11.2%) per 1 MP resolution increase [15], which can be a reference to the effect of the camera resolution for UAV-based photogrammetry as well.

Flight Design: The influential factors in designing a flight include image capturing configurations, camera position, and ground control points. In the image-capturing configurations, the flight path pattern is the most fundamental setting. There are three commonly used patterns for flight path design in building surveys, which are Strip Path [4,34,92], Polygon Path [18,25,44,91,98], and Circle Path [7,45] patterns. As shown in Fig. 4, the strip path pattern means flying a UAV in strips with a zig-zag pattern across the area of interest [24,46,79]. The strip path could be further divided into horizontal (Fig. 4. a) and vertical (Fig. 4. b) paths. Polygon path patterns (Fig. 4. c) set waypoints surrounding the survey target, and the shape of the close-loop generated from the path varies according to the shape of the building envelope, which in most cases are rectangular or polygon shapes [7,18,98]. The circle path pattern (Fig. 4. d) is similar to the polygon pattern because both patterns collect data surrounding the target, with just the shape of the path being different [4,11].

Some studies have combined both the strip path and the polygon path, creating a camera network to capture all angles of buildings [29,82,101]. For example, in the study by Grenzdörffer et al. [29], the authors combined the strip path pattern and the circle path pattern to acquire images from a network of more dense camera positions for a cultural heritage monument's main structure. A similar approach was also adopted by Tran et al., in which a combination of strip and circle path patterns was adopted for the data collection of a school building [87]. Apart from manual flight path design, previous studies have also explored various methods for automatic flight path planning. Tan et al. addressed the UAV coverage path planning problem by employing a genetic algorithm (GA) to minimize the UAV flight length while ensuring a comprehensive and high-quality image data collection [84]. They also introduced a "global-local" adaptive inspection method integrating BIM and wide-angle camera technology for dynamic UAV flight path adjustment, enhancing efficiency, precision, and cost-effectiveness



Fig. 3. Categorization of factors and evaluation metrics in UAV-based photogrammetric as-built survey for buildings.

Range of influential factors for 3D scene reconstruction from previous literature.

Categorization	Factors		Cases from Literature
Data Collection	Equipment Setup	UAV Characteristics	DJI series products (Inspire, Mavic [18,44,53,77], Phantom [6,32,78,82,91], Matrice [90])
			MD4 series [29,34]
			Ascrech Falcon [12,2/,45]
			Autol Euro Nono [07]
		Comero Characteristics	Different Comerce with recolutions from 10Mp [27] to 49 MD [87]
	Flight Design	Flight Path Pattern	Polygon Pattern [18 25 44 51 77 91 98]
	riight Design	right rain rattern	Circle Pattern [7 45]
			Strip Pattern [4 34 92]
			Mixed Pattern [29,82,87,101]
		Photo Overlap	40%-50% [7]
		I.	60%-80% [25,45]
			70%-80% [18,56]
			86% [34]
		Total Number of Photos	Varying from 27 [18] to 2755 [61]
		3D Distance from Target	Horizontal: 5 m [44] – 100 m [18]
			Vertical: 20 m [4] – 100 m [18]
		Oblique Angle of Camera	Varying from 15° to 45° [4,17,87,92]
		Ground Control Points	Varying from 4 [92] to 117 [29]
Data Processing	Software Applications	Open-source Packages	PMVS [45]
			Bundler [12]
			MicMac [4,61]
		Self-developed Algorithm/Tools	Visual SfM [34]
			MAVMAP [27]
			Interactive SfM [12]
		Commercial Tools	Agisoft Photoscan (now Metashape) [4,17,25,29,53,56,61,79,82,92,98]
			Pix4D [4,6,17,37,44,45,61,62,78,79,92,93]
			Recap Photo [18]
			SUI Zepnyr [4]
			SURE [4,29] DesteMadalar [24,61,00]
			Protovioueter [34,01,92]
			Drone Deploy [18]



Fig. 4. Typical pre-flight path planning (a. Strip Pattern (horizontal), b. Strip Pattern (vertical), c. Polygon Pattern, d. Circle Pattern).

[85]. In addition to single-UAV flight path planning, previous studies have also investigated a multi-UAV path planning approach for efficient 3D reconstruction of post-disaster buildings, balancing flight distance and time while avoiding obstacles [63].

Flight planning also includes planning the photo overlaps for robust data collection and the efficiency of the post-flight processing [73]. Photo overlap percentages were reported in ranges from 40% to about 90% [7,34] (Table 3). We did not find a benchmark study of this factor while it has a significant connection with the flight path pattern, and all previous studies have set specific photo overlap percentages for each survey target. The number of overlapping photos has already been identified as a critical factor in terrestrial photogrammetric modeling [15]. However, the total number of photos in the 3D reconstruction of buildings can considerably vary depending on the building type, survey objectives, and level of detail requirements. For example, Djimantoro and Suhardjanto [18] successfully transformed only 27 images into a reconstructed scene of a commercial project for urban planning, while Murtiyoso, et al. [61] used 2755 images for information documentation

of a historical building. For this reason, instead of the total number of photos, the photo overlap percentages would be a more suitable factor for establishing standards in future studies.

For the camera position, researchers have investigated the relative locations of UAVs during data collection, which is represented by the altitude (vertical distance) and the distance (horizontal distance) between UAVs and the respective target surfaces. Adopting the presentation style of summarizing the cases of UAV applications by Rakha and Gorodetsky [73], Fig. 5 demonstrated various vertical and horizontal distances between UAVs and the target from the collected articles. Based on our observations, the altitude varies from 6 m to 100 m depending on the target buildings. The distance may also be adjusted according to the data collection objectives. The distance between the camera and the object determines the ground sampling distance (GSD), which is the distance between the centers of two adjacent pixels in an image. The building surveys with requirements for higher levels of detail need low GSD and shorter distances between the UAV and the target. For building inspection or damage detection, the typical distance has been found to



Fig. 5. The vertical and horizontal distance between UAVs and the building targets in the compiled literature.

be smaller between 1 and 5 m [9,83]. For example, the distance between camera and the target can be between 2 and 5 m in the assessment of post-disaster areas (e.g., post-earthquake building evaluation) [1,47], or less than 1 m in the detection of façade cracks [22]. For 3D reconstruction of buildings, larger distances have been reported, typically more than 10 m, e.g. between 15 and 20 m in the reconstruction of cultural heritages [29]. The oblique angle of the camera is normally set at 45 degrees [4,92], but the specific angle should be determined by other factors, such as 3D distances (horizontal and vertical distances) between the UAV and the target [4]. One previous study has identified that a combination of UAV images taken at 30° and 90° angles is optimal for generating a sufficient number of matching points with high precision [68].

The last factor in data collection is Ground Control Points (GCPs), which are used to guide the flight of the UAVs in covering the areas of interest. The setting of GCPs is for geo-referencing in data collection and data processing. Agüera-Vega, et al. [3] demonstrated the influence of the number of GCPs on the accuracy of UAV-based products through field studies and stated that both the horizontal and vertical accuracy increases with the increasing number of GCPs. A previous study has identified that the number and the distribution of the GCPs can significantly affect the computational cost and the quality of the reconstructed point cloud model [79]. Nevertheless, similar to the number of photos, the settings of GCPs and the number of GCPs can vary across different buildings, so it may not be suitable to set the number of GCPs as a benchmark feature to be applied for general applications.

The influential factors in data collection may be cross-correlated and thus sometimes the modification of one factor would affect the other factors. For example, the change of photo overlap would affect the total number of photos and the time effiency related to the data collection amd processing. Previous studies have not explored the influence of changing these factors with the consideration of cross-correlation, which is what we have investigated in our case study. Details of the findings will follow in the case study section. data is a critical step in building surveys. As indicated in Fig. 3 and Table 3, the processing methods and tools in previous studies can be divided into three groups: 1) self-developed algorithms/tools, 2) open-source packages, and 3) commercial tools. Some studies have focused on developing algorithmic frameworks to create 3D reconstructed scenes. The Scale Invariant Feature Transform (SIFT) for feature matching and interactive Structure from Motion (SfM) techniques for image transformation [12,64] are among the core algorithmic components for this purpose. Others utilized open-source packages like PMVS [45], Bundler [12], and MicMac [4,61] for image processing. However, most of the previous studies have utilized well-established commercial software applications, such as Agisoft Photoscan (now Metashape), Pix4D, or 3Df Zephyr for data processing. It can be seen from Table 3 that Agisoft Photoscan and Pix4D are the two most frequently used commercial tools in the compiled articles.

Some studies have further compared these commercial tools to identify the ones with better performance. As presented in Table 4, previous studies [4,17,61,79,92] have investigated the accuracy of UAV-based building surveys by adjusting critical factors such as data processing software. By comparing the point clouds generated from UAV images using different data processing software tools, they proved that all of the tested software tools could produce building point clouds with high quality (less than 0.05 m difference between TLS and image-based point clouds). Based on these comparisons, Agisoft Photoscan and Pix4D have been shown to have better overall model accuracy among the commonly used commercial tools [17,79].

4.2.2. Categorization of performance evaluation metrics

For performance evaluation metrics, accuracy, time efficiency, and point cloud characteristics (as shown in Table 5) have been commonly used in previous studies. To evaluate the quality of 3D reconstructed scenes, measurements of these as-built scenes need to be compared with actual real-world data. Accuracy reflects the reliability of photogrammetric measurements in applications. As shown in Table 5, many studies utilized comparisons of point clouds obtained from the image-based

4.2.1.2. Processing methods and tools. The processing of UAV-collected

Performance evaluation of UAV-based photogrammetric image processing by different commercial software tools in previous literature.

Performance Evaluation	Commercial Tools									
	Agisoft Photo Scan	Pix4D	3D Zephyr	MicMac	SURE	Context Capture	Photo Modeler	Visual SfM	3Dsurvey	Ref.
Point cloud mean difference compared with	3	4	5	6	25	4	-	5	-	[4]
TLS data (mm)	0	2	_	2	-	-	4	_	-	[<mark>61</mark>]
	8	5	_	-	-	-	-	-	-	[92]
Maximum dimensional difference compared with TLS data (m)	0.01	0.03	-	-	-	-	-	-	0.06	[17]

Table 5

Performance evaluation metrics for UAV-based photogrammetric 3D scene reconstruction in previous literature.

Categorization	Evaluation Factor	References
Accuracy	 Point cloud (image) to point cloud (TLS) comparison (point cloud accuracy) 	[4,7,12,17,25,29,34,53,56,61,79,82,92,101]
	 Dimensional accuracy 	[7,17,37,78,79,92]
Time Efficiency	 Data collection runtime 	[101]
-	 Data processing runtime 	[12,34,101]
Point Clouds	Number of points in the model	[4,25,27,29,34,45,92,101]
	 Point cloud density 	[44,62,92]

models and the point clouds generated from Terrestrial Laser Scanning (TLS) (as the ground truth data). For example, Aicardi et al. [4] evaluated the quality of point clouds by calculating the minimal, maximal, and average point distances and compared them with TLS-based point clouds. Some studies have also used dimensional accuracy as an evaluation metric. Vacca et al. [92] evaluated the dimensional accuracy of models by comparing the measurements of key dimensions between UAV-based and TLS-based point clouds. Time efficiency is also a performance indicator for UAV photogrammetric surveys. The on-site data collection duration and the post-flight model processing time should be recorded as the reference for evaluation of the time efficiency [15]. As an example, Zheng et al. [101] have demonstrated that the usage of multiple UAVs can dramatically decrease the data collection time from 42 min to 15 min. Another factor that can affect the time efficiency is the control of the UAV during on-site data collection. Compared with manual-flying control, autonomous control with pre-programmed routes is much faster and more efficient. As the third group of metrics, the total number of points in a model and the density of the point clouds have been also considered for performance evaluation. A denser point cloud could provide information with a higher level of detail [25].

4.3. Interrelationships between factors

To analyze the interrelationships between the above-mentioned parameters (influential factors in 3D scene reconstruction and performance evaluation metrics), as noted, we utilized the Social Network Analysis (SNA) method. SNA enables the analysis and visualization of interrelationships between multiple entities in a network using graph theory [55]. For each reviewed article (case) in the selected pool from Table 2, the terms, associated with our parameters of interest, that appeared in each article were counted, while the non-existing terms were set to zero. A two-mode (bi-partite) network, which quantifies the occurrence of each parameter in each paper, was generated. The two-mode network can then be transformed into a one-mode network for the studied parameters to quantify the co-occurrence frequency among all parameters. The heatmap shown in Fig. 6 presents the calculated probability of the co-occurrence for each set of two factors in the network. The values in each cell represent the frequency of co-occurrence of two factors in the reviewed literature. The higher the value, the higher the co-occurrence frequency of the two factors in previous studies.

The parameters co-occurrence matrix was used to measure the centrality of each parameter (factor) using the software Netminer [76]. As shown in Fig. 7, the size of each node in the SNA network represents the degree of centrality of each factor, which identifies the most important or influential factors. It can be observed that the most influential factors during the data collection phase (blue nodes) show very high degrees of centrality, especially the factors representing equipment setup configurations such as flight path, and the total number of photos. For postflight data processing factors (green nodes), commercial software tools were most commonly used. Performance evaluation metrics (red nodes) of accuracy and point cloud quality were broadly investigated to evaluate the performance of 3D model reconstruction. Meanwhile, the thickness of each link in the SNA network represents the frequency of cooccurrence of two correlated factors in the reviewed literature. The thicker the link, the higher the frequency of co-occurrence between the two factors. The equipment setup, flight path, distance, number of photos, and commercial software/tools are the important factors that have been frequently mentioned when studying the accuracy performance of the reconstructed models. For the point cloud quality, equipment setup, flight path, distance, number of GCPs, and the number of photos were mostly related. However, time efficiency was seldom mentioned in the collected studies because it was seldom used in evaluations. Based on the SNA analysis, it can be concluded that the equipment setup, flight path patterns, the 3D (horizontal and vertical) distance between the UAV and the target, the overlap between photos, the number of photos, and the choice of the data processing commercial software may play a critical role in the quality of the 3D reconstructed as-built scenes.

The majority of the selected studies have investigated the 3D scene reconstruction with specific pre-flight settings and performance evaluation metrics, which makes it difficult to evaluate the impact of the influential factor ranges on the outcome. Although we sought to quantify the impact of influential factors on performance indicators using the selected literature, the numeric patterns for each parameter (e.g., photo overlap percentages, photo numbers, and distances) are difficult to be synthesized and analyzed. For this reason, in Section 5 we designed a case study to investigate the impact of three identified important factors (i.e., flight path pattern, the distance between the UAV and the target, and photo overlap) on three frequently used performance measurements including dimensional accuracy, time efficiency, and point cloud quality.



Fig. 6. The heatmap of the co-appearance probability of the identified factors in the compiled literature.



Fig. 7. Degree centrality of and potential interrelationships between different identified parameters.

5. Case study

5.1. Design of experiment for factor assessment study

For the case study of factor assessment, we selected a mixed-use building on Virginia Tech's campus as the target object. The building has a regular cuboid shape, and the overall dimensions of the building are 37 m \times 17.5 m \times 20 m (length x width x height). For equipment

setup, considering their wide application in academic research and the market penetration of DJI products, we chose a DJI Mavic Mini drone equipped with a 12 MP camera for our experiments. The comparison of different camera resolutions was not conducted in this case study as previous studies have already identified the influence of camera resolution on the photogrammetry outcome [15]. Several data-processing software tools have also been utilized in previous studies [4]. Through a comparison of reported quality from various software solutions in the

literature (shown in Table 4), Agisoft Photoscan (now Metashape) was selected for processing. We collected the data on cloudy days with an overcast sky and wind speed in the range of 3–6 km/h to mitigate the effects of interfering environmental factors. The following ranges of factors were used in our data collection:

- *Flight path:* we set three flight path types for the building survey, consisting of a horizontal strip path pattern, a polygon path pattern, and a circle path pattern (Fig. 8. a-d, e, and f.)
- *Horizontal distance*: A close distance of 15 m and a long distance of 30 m were used (Fig. 8. a, d). For polygon and circle path patterns, the distance values were higher (at 30–50 m).
- *Vertical distance:* The altitude of flight paths above the top of the building was set as 30 m, which was 10 m above the roof of the building.
- *Photo overlap percentages*: 80%, 60%, and 40% were used (Fig. 8. a, b, c).

Apart from using the collected UAV-based data for 3D reconstruction, a terrestrial laser scanner (TLS) was also utilized to create point clouds of the building as a baseline for performance evaluations. Previous studies have broadly utilized TLS-based point clouds and models as the benchmark for model evaluation [4,61,92], and it has been universally acknowledged that laser scanners can be used to rapidly collect data with high accuracy. A FARO Focus laser scanner was utilized with the predefined scan profiles of "Outdoor HDR", the resolution of 1/8, and the quality of " $4\times$ ". The accompanied FARO SCENE software was used for laser scanner data processing. Three stations on each side of the building (12 stations in total) were set for laser scanning (Fig. 9), and the model performance was evaluated following the same criteria for the UAV-based models.

Upon developing the 3D as-built point clouds, the large-scale building dimensions (i.e., building length, width, and height) and selected detailed dimensions (e.g., the width of a window, the height of a door) were measured and compared with the dimensions of as-built drawings of the building (ground truth data). Errors and error rates were calculated to measure the dimensional accuracy. To evaluate the point cloud quality, the number of points in the entire point cloud and the number of points on different surfaces were measured. During the data collection and post-processing, the time needed for data collection and processing was recorded for time efficiency comparisons.

5.2. Findings from the case study

During flight planning, we utilized UgCS, which is a commercial software tool with various UAV toolsets for land surveying and industrial inspections. In consideration of safety issues such as avoiding surrounding buildings and pedestrians, we identified reference points for the building and then collected data with manual control. Therefore, while we set the photo overlap as a fixed value in the design, the actual photo overlap may vary by $\pm 5\%$ around the set values as shown in Table 6. Following data collection, we developed 3D reconstructed scenes in the form of point clouds and built mesh models of the building with Agisoft Photoscan software. In the remainder of the text, we also refer to these as-built scenes as models.

As shown in Table 6, seven 3D models with different factor configurations were built to compare the influence of various factors. Fig. 10 shows the oblique view of these models. These visualizations show clear quality differences between the models. Among all cases, six proper models were developed, while for model c (Fig. 10.c) the images were not properly aligned to successfully form the complete building mesh model. For model c, 60 photos were collected as the original setting, while only 33 of them were properly aligned by the software despite different rounds of data collection. Additionally, as the laser scanner only collected terrestrial data, the top surface was not modeled in the TLS-based model (Fig. 10. g) compared with other UAV-based models with sufficient information on the building's top surface. Apart from that, there are also other differences among these seven models. Further quantitative comparisons have been discussed in the following subsections.



Fig. 8. Pre-flight path planning for the case study building (a. Strip Pattern (80% photo overlap), b. Strip Pattern (60% photo overlap), c. Strip Pattern (40% photo overlap), d. Strip Pattern (30 m), e. Polygon Pattern, f. Circle Pattern).



Fig. 9. Terrestrial laser scanner data collection stations (12 stations in total).

Table 6 Seven 3D as-built reconstruct

Seven	3D	as-built	reconstructed	scenes	and	their	mesh	models	with	various
factor	sett	ings.								

Models	Flight Path	Photo Overlap	Horizontal Distance (meters)	Number of Photos
a (Strip – 80% -15)	Strip Pattern	80% (75%– 85%)	15	240
b (Strip – 60% -15)	Strip Pattern	60% (55%– 65%)	15	115
c (Strip – 40% -15)	Strip Pattern	40% (35%– 45%)	15	60 (33 aligned)
d (Strip – 80% -30)	Strip Pattern	80% (75%– 85%)	30	82
e (Polygon – 80%)	Polygon Pattern	80% (75%– 85%)	20–30	60
f (Circle – 80%)	Circle Pattern	80% (75%– 85%)	40–50	82
g (Laser Scanner)	N/A	N/A	15	N/A

5.2.1. Accuracy analysis

We imported the reconstructed 3D models from Photoscan into Autodesk Revit to measure the dimensions of the 3D scenes. For each model, we collected six major dimensions such as the length, width, and height of the building (Fig. 11. a), and extracted six detailed dimensions, such as the width and height from a window on the east side wall and a door of the main entrance in the building (Fig. 11. b). These measurements of the reconstructed 3D models were then compared with the ground truth dimensions to evaluate the accuracy of the reconstructed 3D models. The accuracy analysis was conducted by calculating the average absolute error, average error rate, and standard deviation of these measurements.

Table 7 presents the results of the dimensional accuracy assessments. Note that all values in the table are mean values of the related items. The average absolute errors and error rates are the mean values calculated from the absolute errors and error rates of measured dimensions, respectively. The observations from Table 7 have been summarized below and the error rate comparison has been presented in Fig. 12.

As shown in Fig. 12, the strip path pattern (Table 7. a) demonstrates superior performance compared with the polygon path (Table 7. e) and the circle path (Table 7. f) patterns, particularly in measuring detailed dimensions, where the average error rate of all four strip path models is

around 1%, while the average error rate of the other two path types (polygon and circle) are as high as 6%. As expected, the decrease in the photo overlap leads to a decrease in dimensional accuracy. When assessing all dimensions, the error, and the standard deviation gradually increase as the photo overlap decreases from 80% to 40% (Table 7. a, b, and c). Compared with the model generated from the laser scanner (Table 7. g), UAV-based models exhibit a considerably higher standard deviation of the measured absolute errors. In Fig. 12, when considering the error rate, the small gap between the UAV-based (strip pattern with 80% photo overlap) model and the TLS-based model indicates the high accuracy that the UAV-based models can reach.

Table 7 also shows that the impact of the change in photo overlap on the major dimensions is larger than its impact on detailed dimensions. By comparing the variations between 80% and 40% photo overlaps, we can see that the increase in the error rates of the detailed dimensions is smaller than the increase of error rates in major dimensions as the photo overlap decreases. The effect of changing the distance between the UAV and the building facade is relatively less considerable than that of the photo overlap. When comparing the impact of distance and photo overlap, the increase of error rate between the model with a 15-m (Table 7. a) distance and the model with a 30-m distance (Table 7. d) (0.08%) is less than the difference between model built with 80% photo overlap (Table 7. a) and the model built with 60% photo overlap (Table 7. b) (0.15%).

5.2.2. Point clouds (model) quality analysis

Based on performance evaluations reported in previous studies [4,18,24,58], we defined the total number of points in the reconstructed scenes and the density of points as reference metrics for quality assessments. To calculate point cloud density, we segmented the point cloud into three surfaces (South, West, and Top) and two detailed components (a Window and Entrance Door) as presented in Table 8. The point cloud density was calculated based on the number of points in the related facades. The density of the top surface is higher than the others since the top surface also includes components, such as building systems and parapet walls. The point cloud density of the windows and doors (containing large portions of glass) is generally lower than the other parts of the building surfaces (containing concrete or steel material).

As presented in Table 8 and Fig. 13, with the same photo overlap (80%), the number of points in the scenes from strip patterns (Table 8. a)



Fig. 10. 3D as-built scenes (i.e., mesh models) reconstructed based on different flight paths (a. Strip Pattern (80% photo overlap), b. Strip Pattern (60% photo overlap), c. Strip Pattern (40% photo overlap), d. Strip Pattern (30 m), e. Polygon Pattern, f. Circle Pattern, g. Terrestrial Laser Scanner).



Fig. 11. Dimension references of the case study building (Major Dimensions and Detail Dimensions) from a 3D model of the building - the dimensions reflect the actual as-built dimensions of the building.

is larger than the number of points in those from polygon patterns (Table 8. e) and the circle pattern (Table 8. f). The number of points in the point cloud decreases as the photo overlap percentage decreases or the distance between UAVs and the building facade increases (Table 8. a-c). Nevertheless, in contrast to the dimensional accuracy, the impact of changing the object distance is larger than changing photo overlap percentages when it comes to point cloud density. The decrease in point cloud density between a reconstructed scene from images taken at a 15-

m distance (Table 8. a) and at a 30-m distance (Table 8. d) is much larger than the decrease of point cloud density between an as-built scene from 80% photo overlap (Table 8. a) and from 60% photo overlap (Table 8. b). The low point cloud density of the circle pattern (taken at 40–50 m from the building facade) could also be an indicator of the significant impact the distance has on the point cloud density. For non-strip patterns, larger distances are required so that the whole surface of a facade can be captured, which is also a reason for the drop in the point cloud

Comparison	of dimensiona	l accuracy fo	r different	reconstructed	scenes and	their mesh	models.

Model	All Dimensions			Major Dimensions			Detailed Dimensions		
	Abs. Error (cm)	Error Rate	STD	Abs. Error (cm)	Error Rate	STD	Abs. Error (cm)	Error Rate	STD
a (Strip – 80% -15)	6.15	0.72%	26.57	11.00	1.0%	34.06	1.27	0.45%	5.41
b (Strip – 60% -15)	8.99	0.87%	36.75	15.24	0.68%	48.44	2.74	1.05%	5.16
c (Strip – 40% -15)	17.65	1.48%	71.40	30.48	1.76%	85.85	2.29	1.13%	5.74
d (Strip – 80% -30)	4.70	0.80%	13.08	7.62	0.95%	14.83	1.80	0.65%	5.03
e (Polygon – 80%)	50.80	6.12%	165.28	84.25	6.08%	202.31	17.35	6.15%	43.08
f (Circle – 80%)	14.53	3.07%	42.37	12.19	1.03%	17.27	17.48	5.63%	65.28
g (Laser Scanner)	3.25	0.5%	4.05	6.00	0.9%	4.05	0.5	0.1%	1.22





Table 8

Comparison of point cloud characteristics in different reconstructed scenes.

Model	Number of Points in the Whole Model	Point Cloud Density (pts/m ²)					
		South Facade	West Facade	Top Surface	Window	Door	
a (Strip – 80% -15)	117,074,368	26,198	26,909	40,061	23,279	15,220	
b (Strip – 60% -15)	83,803,128	22,414	21,511	33,049	20,545	12,722	
c (Strip – 40% -15)	27,296,209	10,208	1262	12,522	8687	5083	
d (Strip – 80% -30)	51,037,110	8149	7797	12,641	7479	5820	
e (Polygon – 80%)	42,487,232	7264	6354	3153	6478	5456	
f (Circle – 80%)	12,338,180	473	521	693	642	318	
g (Laser Scanner)	50,024,868	19,758	5995	N/A	4810	8318	



Fig. 13. Comparison of point cloud density for selected surfaces across different reconstructed scenes.

density.

The TLS-based point cloud (Table 8. g) did not include information on the building's top surface, resulting in a lower number of points. Also, the point cloud density was not the best TLS-based point cloud attribute given that we did not use the highest resolution and quality settings, due to the consideration of time efficiency, which will be further discussed in the next section. The differences in the point cloud quality are apparent in Fig. 14–16, which show comparisons of the South facade and the main entrance of the case study building. It can be seen from the figures that point cloud density is reflected in the quality of the mesh models and the level of detail. In Fig. 15 a, b, c, and Fig. 16, we can still see the incised building name and the basic frame of the building entrance door, while in Fig. 15 d, e, and f, this information is mostly lost.

5.2.3. Time efficiency analysis

We recorded the elapsed times for both data collection and data processing to analyze the time efficiency of the 3D scene reconstruction process. Table 9 presents the results, which show that the data processing time accounts for a major portion of the overall processing time. While hardware characteristics such as differences in graphics processing units (GPU) can also impact processing time, we focused solely on the impact of pre-flight design factors on time efficiency. The data processing was conducted on the same device with a Core i7 processor and 16 GB of RAM to avoid the impact of hardware differences.

The results presented in Table 9 indicate that the total processing time for strip patterns is much higher than that for circle and polygon patterns. The model from the strip pattern with 40% photo overlap is an outlier case, as the model was not properly aligned in the 3D reconstruction process. Generally, it can be observed that as the photo overlap decreases or the distance increases, the data processing time decreases. One of the major reasons for the difference is the number of photos that need to be aligned. The mesh model from the strip pattern with 80% photo overlap covers 240 photos in total, and the data processing time (49.8 h) is much longer than the data processing time of any of the other mesh models, which only cover 60-115 photos. In the case of terrestrial laser scanning, most of the time was spent on data collection. A single scan can take up to 2 h when using the highest resolution and quality settings. However, with 12 scanning stations and the practical requirements in real-world scenarios, we used the FARO's predefined scan profile of "Outdoor HDR" that limited the time of each full scan to 15 min. Compared with laser scanning, UAV-based data collection takes considerably less time to finish, but this may come at the cost of longer data processing time and reduced model performance, which will be discussed in the next section.

5.2.4. Trade-off between as-built reconstructed scene quality and time efficiency

The case study results suggest that the 3D scene reconstructed by images collected using the strip path pattern outperforms polygon and circle path patterns in terms of dimensional accuracy and point cloud density. However, better model performances also require more postprocessing time because larger photo overlaps result in greater numbers of photos. Changing the distance between the UAV and the target building has a limited impact on the dimensional accuracy but a considerable effect on the point cloud density. On the contrary, the photo overlap percentage plays a crucial role in model performances. When the photo overlap was set to 40%, photos of some surfaces could not be properly aligned in this case study, and the 3D scene could not be properly formed. As the photo overlap percentage increases, it leads to higher dimensional accuracy and point cloud density, but this also results in an increase in total post-processing time. As such, there is a trade-off between the quality of the reconstructed scene (dimensional accuracy, point cloud density) and the time efficiency (total time) as illustrated in Fig. 17.

As shown in Fig. 17, for photo overlap percentages below 60%, increasing the percentage considerably affects the quality of 3D reconstructed scenes, with a relatively high variation rate for both dimensional accuracy and point cloud density. The decrease in error rate and increase in point cloud density change dramatically when moving from 40% to 60% overlaps. However, once the photo overlap percentage goes above 60%, the rate of improvement in the quality drops considerably. The photo overlap percentage of around 60% (50%-60%) acts as a watershed of allocative efficiency in this case study experiment. It is efficient to increase the photo overlap when it is under 60%, while the efficiency cost of processing will be higher than the benefits of model improvement when the overlap percentage is above 60%. Nevertheless, it is worth noting that the observed allocative efficiency point could be a feature of this case study. For generalization, more applied cases with different influential factors and various building types need to be investigated in future studies.

5.2.5. Other influential factors

In addition to the flight design factors discussed earlier, we identified two other factors that may affect the performance of UAV-based 3D



Fig. 14. Point cloud of the South facade of the case study building (a. Strip Pattern (80% photo overlap), b. Strip Pattern (60% photo overlap), c. Strip Pattern (40% photo overlap), d. Strip Pattern (30 m), e. Polygon Pattern, f. Circle Pattern).

T. He et al.

Automation in Construction 161 (2024) 105323



Fig. 15. Point cloud of the main entrance door of the case study building (a. Strip Pattern (80% photo overlap), b. Strip Pattern (60% photo overlap), c. Strip Pattern (40% photo overlap), d. Strip Pattern (30 m), e. Polygon Pattern, f. Circle Pattern).



Fig. 16. Details of the terrestrial laser scanner-based Point cloud (Left: south façade; Right: main entrance door).

Table 9

Time	elapsed	for processing	images	to 3D	reconstructed	scenes	and	their	mesh
mode	ls.								

Model	Data Collection (h)	Data Processing (h)	Total Time (h)
a (Strip – 80% -15)	0.8	49.8	50.7
b (Strip – 60% -15)	0.4	17.2	17.6
c (Strip – 40% -15)	0.2	1.3	1.5
d (Strip – 80% -30)	0.3	7.5	7.9
e (Polygon – 80%)	0.2	3.3	3.5
f (Circle – 80%)	0.2	2.3	2.4
g (Laser Scanner)	3.5	1	4.5

reconstruction in the case study trial tests. These factors are weather conditions during data collection and the collection of data from the intersection between two surfaces of the target building. Weather conditions can considerably impact the quality of the collected data. Data collection on a sunny day can lead to data processing errors. The reflection of sunlight from glass surfaces, the shadows cast by the sunlight, and the illumination differences can be sources of errors. The influence of lighting and weather conditions on UAV photogrammetry has also been reported in previous literature, in which they also collect data during an overcast period to avoid the negative effect of the changing illumination [76]. Therefore, it is recommended to collect data on cloudy days with an overcast sky. Collecting data from the point of convergence between two distinct building surfaces, such as the south facade and the roof, can pose a challenge due to a misalignment. This misalignment can, in turn, lead to the failure of image matching during the data processing stage. To circumvent these challenges, it is recommended to incorporate a strategic flight path plan that includes coverage of the interface regions where two surfaces meet. In the planning of the UAV's flight path, it is important to ensure that the UAV camera is oriented at a 45-degree angle along each edge, thereby ensuring unobstructed visibility of both adjoining building surfaces. Employing this approach in data collection can markedly mitigate the occurrence of image misalignment issues during data processing and thereby enhance the overall quality of the resulting 3D model.



Fig. 17. Comparison of the 3D model reconstruction performance for different photo overlap percentage using strip pattern for flight path: a. Dimensional accuracy vs. Total time, b. Point cloud density vs. Total time).

6. Discussion

6.1. Data Schema for building survey based on UAV-based photogrammetry

As we evaluated the selected literature, it was observed that these studies did not uniformly specify the detailed information on influential factors as identified in this article. This could pose a challenge in synthesizing the information across multiple studies to draw generalized conclusions. Accordingly, we propose a generalizable data schema that could be adopted in future studies and practices in the field of UAV-based photogrammetry. In this way, the findings in individual studies can be analyzed and compared across studies. We have presented the data requirements for such a schema in Table 10. If future researchers or industry practitioners include these data items in reporting their findings, a larger-scale meta-analysis and comparative assessment of modeling performance could be facilitated, contributing to the development of standardized procedures for UAV-based photogrammetry data collection and processing for buildings' as-built surveys.

As shown in the proposed data schema, various factors can affect the UAV-based 3D as-built scene reconstruction, including survey target shapes and dimensions, data collection-related features, and data processing methods. Different evaluation criteria and point cloud measurements should also be considered. In previous studies, researchers attempted to achieve the optimal performance quality of the reconstructed models in their case studies. However, different objectives and levels of detail for different applications may require different configurations to improve time efficiency. While some studies have investigated the impact of specific factors on the model performance individually, few have investigated the collective effect of these factors. Although we conducted a case study to investigate the trade-off between different factors and reconstructed model quality, several limitations still exist that require further investigation in future studies.

6.2. Limitations and future studies

In this study, a detailed factor assessment was conducted through a systematic literature review and a case study field evaluation. While the outcome of this endeavor offers valuable insights, there are inherent limitations of the case study conducted in this research. First, our case study was restricted in scope, assessing a limited subset of influential factors with relatively modest variations. The selection of photo overlap percentages (40%, 60%, 80%) and the two selected fixed UAV-to-building distances (15 m and 30 m) presents an opportunity for more

comprehensive and diversified experimentation. Given the computingintensive nature of the analyses, it is challenging to compile an extensive repository of sample data, encompassing diverse factor combinations, distinct building typologies, and a broad array of architectural forms. Deploying cloud computing resources, a future avenue of research could involve a broader spectrum of factors, including additional photo overlap percentages, broader variations in the number of photos, investigated distances, altitudes, and ground control points (GCPs), as well as comparisons between different processing software solutions, and utilized hardware specifications. Additionally, the inclusion of various building types, sizes, and real-world applications can help expand the foundation developed in this study.

Furthermore, while terrestrial laser scanning served as the ground truth data source for comparison in this study, comparison against alternative means of data collection, such as terrestrial robots (e.g., agile mobile robots [39]), remains to be explored [96]. Future investigations may encompass a more extensive examination of the disparities between 3D as-built reconstructions derived from UAV-based photos and those gathered by alternative data collection methods, such as alternative ground-based methods. Future studies should also examine the synergistic potential of integrating UAV and terrestrial photogrammetry or laser scanning data to enhance model accuracy [41,74,90]. Empirical studies have proposed frameworks for operating mobile robots equipped with laser scanning systems in cluttered outdoor environments with the aid of UAVs [40,69], and developed methods to automatically register the 3D point clouds collected both from UAVs and tresterial robots [68]. Such integrated approaches hold promise in pushing the boundaries of precision in building surveys and construction analysis.

7. Conclusion

The practices of UAV-based as-built scene reconstruction have achieved significant advancements, primarily focusing on optimizing the quality of image-based 3D reconstructed scenes. However, in the literature, there exists a gap in the systematic identification and analysis of critical factors in UAV-based photogrammetry that could have a significant impact on performance. Therefore, in this study, we presented a comprehensive factor assessment using a systematic literature review, through which a categorization of relevant parameters was developed for the three phases of scene reconstruction: (1) flight design for data collection, (2) data processing, and (3) performance evaluation. Through social network analysis and frequency assessment in the selected literature, we identified the most influential factors to be flight path patterns, the distance between the UAV and the target building,

Data requirements for a data schema in UAV-based 3D as-built scene reconstruction.

Section	Factor	Data/Information
Survey Target	Target Information	 Target Type (e.g., residential building, commercial building, or cultural heritage) Target general information (including (1) shape: cuboid, tower, ring shape, L shape, etc.; (2) number of floors; (3) surrounding context (e.g., obstacles)) Target dimensions (length, width, hericht)
Data Collection	Equipment Setup	 UAV type and make UAV specifications (flight duration time, flight speed, and whether they support autopilot or not) Attached camera type and make Attached camera specifications (mostifically, comera specifications)
	Flight Design	 Flight path (strip pattern, polygon pattern, circle pattern, or mixed pattern)
	Spatial and Temporal	 Camera positions (including (1) distance between the camera and the target; (2) altitude of the camera; (3) oblique angle of the camera) Image overlap percentage (from 0% to 100%) Ground control points specification (whether GCPs were set or not; the number of GCPs; positions of the GCPs) Number of images that were planned to be collected Climate condition at the time of data collection (e.g., sunlight condition:
	Information	sunny sky or overcast sky; wind speed)
Data Processing (Model Building)	Processing software information Model building information	 The type, name, and version of the data processing tools/software (commercial product, open-source package, or self-developed system) Number of aligned images in creating the point cloud/ model
Performance Evaluation	Evaluation Criteria	 Lapsed time for data processing Accuracy evaluation type (dimensional accuracy or point cloud to point cloud (TLS) deviation) Model accuracy metrics (e.g., mean dimension errors, mean point distance from TLS point cloud)
	Point Cloud	 Number of points in the model Point cloud density specifications (e. g., point cloud density of specific facade)

and photo overlap percentages. Subsequently, we conducted an in-depth case study to quantitively analyze the trade-off between these factors and their impact on the quality of the reconstructed scenes. Performance evaluations revealed that the strip pattern flight path results in higherquality point clouds when compared to polygon and circle patterns. The distance between the UAV and the target building has a more profound influence on point cloud density compared to the dimensional accuracy. Among all factors examined, photo overlap percentages emerged as the primary driver of 3D scene reconstruction performance. We identified a trade-off between the point cloud quality and time efficiency, where a photo overlap percentage of about 60% can be the tipping point of allocative efficiency in the case study experiment. For detailed building elements, a strip pattern with a close-range (about 15 m) data collection process resulted in high point cloud density and dimensional accuracy. The comparisons between UAV-based models and terrestrial laser scanner-based models revealed that UAV-based photogrammetry (using a strip pattern with 80% photo overlap) can considerably reduce data collection time while achieving equivalent model quality compared to laser scanning.

This study contributes by systematically identifying and quantitatively analyzing factors and metrics in UAV-based photogrammetry, expanding practical knowledge for as-built building surveys. The findings could offer guidance for efficient and precise as-built building surveys with standardized procedures, which in turn could benefit industry practices and future research in this area. Moreover, future studies can leverage the presented categorization and suggested data requirement schema to move towards standardized procedures that could balance the trade-off between data collection and processing efforts and the reconstructed scene quality. Additionally, the findings could contribute to the automation (with pre-set factors and survey objectives) of larger-scale UAV-based photogrammetry workflows that could in turn support the creation of digital twins at different scales.

Declaration of generative AI in the writing process

During the preparation of this work, the authors used ChatGPT in order to refine and polish parts of the text. After using this tool/service, the authors reviewed and edited the content as needed, and take full responsibility for the content of the publication.

CRediT authorship contribution statement

Tianzhi He: Conceptualization, Data curation, Investigation, Methodology, Visualization, Writing – original draft. **Kaiwen Chen:** Investigation, Methodology, Writing – original draft, Writing – review & editing. **Farrokh Jazizadeh:** Conceptualization, Investigation, Resources, Supervision, Writing – original draft, Writing – review & editing. **Georg Reichard:** Resources, Supervision, Writing – review & editing.

Declaration of competing interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

Data availability

The data cannot be shared.

References

- S.M. Adams, M.L. Levitan, C.J. Friedland, High resolution imagery collection utilizing unmanned aerial vehicles (UAVs) for post-disaster studies, Adv. Hurricane Eng. Learn. Our Past (2013) 777–793, https://doi.org/10.1061/ 9780784412626.067.
- [2] B. Adebiyet, F. Iliuf, E. Orynbassarova, A. Chernov, K. Idrissov, 3D modeling of Satbayev university based on laser scanning and UAV data, Int. Arch. Photogramm. Remote. Sens. Spat. Inf. Sci. 46 (2022) 1–6, https://doi.org/ 10.5194/isprs-archives-XLVI-5-W1-2022-1-2022.
- [3] F. Agüera-Vega, F. Carvajal-Ramírez, P. Martínez-Carricondo, Assessment of photogrammetric mapping accuracy based on variation ground control points number using unmanned aerial vehicle, Measurement 98 (2017) 221–227, https://doi.org/10.1016/j.measurement.2016.12.002.
- [4] I. Aicardi, F. Chiabrando, N. Grasso, A.M. Lingua, F. Noardo, A. Spanò, UAV photogrammetry with oblique images: first analysis on data acquisition and processing, international archives of the photogrammetry, Remote Sen. Spat. Inf. Sci. 41 (2016), https://doi.org/10.5194/isprs-archives-XLI-B1-835-2016.
- [5] G. Albeaino, M. Gheisari, B.W. Franz, A systematic review of unmanned aerial vehicle application areas and technologies in the AEC domain, J. Inf. Technol. Constr. 24 (2019) 381–405. http://www.itcon.org/2019/20.
- [6] T. Bakirman, B. Bayram, B. Akpinar, M.F. Karabulut, O.C. Bayrak, A. Yigitoglu, D. Z. Seker, Implementation of ultra-light UAV systems for cultural heritage documentation, J. Cult. Herit. 44 (2020) 174–184, https://doi.org/10.1016/j. culher.2020.01.006.

- [7] T. Bertram, T. Bock, A. Bulgakov, A. Evgenov, Generation the 3D model building by using the quadcopter, ISARC, in: Proceedings of the International Symposium on Automation and Robotics in Construction Vol. 31, IAARC Publications, 2014, pp. 768–773, https://doi.org/10.22260/ISARC2014/0105.
- [8] A.Y. Chen, Y.-N. Huang, J.-Y. Han, S.-C.J. Kang, A review of rotorcraft unmanned aerial vehicle (UAV) developments and applications in civil engineering, Smart Struct. Syst. 13 (6) (2014) 1065–1094, https://doi.org/10.12989/ sss.2014.13.6.1065.
- [9] K. Chen, G. Reichard, A. Akanmu, X. Xu, Geo-registering UAV-captured closerange images to GIS-based spatial model for building façade inspections, Autom. Constr. 122 (2021) 103503, https://doi.org/10.1016/j.autcon.2020.103503.
- [10] Z. Chen, W. Zhang, R. Huang, Z. Dong, C. Chen, L. Jiang, H. Wang, 3D modelbased terrestrial laser scanning (TLS) observation network planning for largescale building facades, Autom. Constr. 144 (2022) 104594, https://doi.org/ 10.1016/j.autcon.2022.104594.
- [11] E. Christy, R.P. Astuti, B. Syihabuddin, B. Narottama, O. Rhesa, F. Rachmawati, Optimum UAV flying path for device-to-device communications in disaster area, in: 2017 International Conference on Signals and Systems (ICSigSys), IEEE, 2017, pp. 318–322, https://doi.org/10.1109/ICSIGSYS.2017.7967064.
- [12] S. Daftry, C. Hoppe, H. Bischof, Building with drones: Accurate 3D facade reconstruction using MAVs, in: 2015 IEEE International Conference on Robotics and Automation (ICRA), IEEE, 2015, pp. 3487–3494, https://doi.org/10.1109/ ICRA.2015.7139681.
- [13] F. Dai, Y. Feng, R. Hough, Photogrammetric error sources and impacts on modeling and surveying in construction engineering applications, Vis. Eng. 2 (1) (2014) 2, https://doi.org/10.1186/2213-7459-2-2.
- [14] F. Dai, M. Lu, Assessing the accuracy of applying photogrammetry to take geometric measurements on building products, J. Constr. Eng. Manag. 136 (2) (2010) 242–250, https://doi.org/10.1061/(ASCE)CO.1943-7862.0000114.
- [15] F. Dai, A. Rashidi, I. Brilakis, P. Vela, Comparison of image-based and time-of-flight-based technologies for three-dimensional reconstruction of infrastructure, J. Constr. Eng. Manag. 139 (1) (2013) 69–79, https://doi.org/10.1061/(ASCE) CO.1943-7862.0000565.
- [16] S. Dastgheibifard, M. Asnafi, A review on potential applications of unmanned aerial vehicle for construction industry, Sustain. Struct. Mater. 1 (2) (2018) 44–53, https://doi.org/10.26392/SSM.2018.01.02.044.
- [17] S.I. Deliry, U. Avdan, Accuracy evaluation of UAS photogrammetry and structure from motion in 3D modeling and volumetric calculations, J. Appl. Remote. Sens. 17 (2) (2023) 024515, https://doi.org/10.1117/1.JRS.17.024515.
- [18] M.I. Djimantoro, G. Suhardjanto, The advantage by using low-altitude UAV for sustainable urban development control, in: IOP Conference Series: Earth and Environmental Science vol. 109, IOP Publishing, 2017, pp. 012–014, https://doi. org/10.1088/1755-1315/109/1/012014.
- [19] Q.F. Dupont, D.K. Chua, A. Tashrif, E.L. Abbott, Potential applications of UAV along the construction's value chain, Proc. Eng. 182 (2017) 165–173, https://doi. org/10.1016/j.proeng.2017.03.155.
- [20] H. Eisenbeiß, UAV photogrammetry, ETH Zurich (2009), https://doi.org/ 10.3929/ethz-a-005939264. Accessed date: July 2022.
- [21] F. Elghaish, S. Matarneh, S. Talebi, M. Kagioglou, M.R. Hosseini, S. Abrishami, Toward digitalization in the construction industry with immersive and drones technologies: a critical literature review, Smart Sustain. Built Environ. (2020), https://doi.org/10.1108/SASBE-06-2020-0077.
- [22] A. Ellenberg, A. Kontsos, I. Bartoli, A. Pradhan, Masonry crack detection application of an unmanned aerial vehicle, Comput. Civil Build. Eng. 2014 (2014) 1788–1795, https://doi.org/10.1061/9780784413616.222.
- [23] C. Eschmann, C.-M. Kuo, C.-H. Kuo, C. Boller, Unmanned aircraft systems for remote building inspection and monitoring, in: 6th European Workshop on Structural Health Monitoring (EWSHM 2012), Dresden, Germany, 2012. https:// www.ndt.net/?id=14139.
- [24] C. Eschmann, T. Wundsam, Web-based georeferenced 3D inspection and monitoring of bridges with unmanned aircraft systems, J. Surv. Eng. 143 (3) (2017) 04017003, https://doi.org/10.1061/(ASCE)SU.1943-5428.0000221.
- [25] S. Espositoa, P. Fallavollitaa, W. Wahbehb, C. Nardinocchic, M. Balsia, Performance evaluation of UAV photogrammetric 3D reconstruction, in: 2014 IEEE Geoscience and Remote Sensing Symposium, IEEE, 2014, pp. 4788–4791, https://doi.org/10.1109/IGARSS.2014.6947565.
- [26] X. Feifei, L. Zongjian, G. Dezhu, L. Hua, Study on construction of 3D building based on UAV images, Int. Arch. Photogramm. Remote. Sens. Spat. Inf. Sci. 39 (2012) B1, https://doi.org/10.5194/isprsarchives-XXXIX-B1-469-2012.
- [27] F. Fraundorfer, Building and site reconstruction from small scale unmanned aerial vehicles (UAV's), in: 2015 Joint Urban Remote Sensing Event (JURSE), IEEE, 2015, pp. 1–4, https://doi.org/10.1109/JURSE.2015.7120481.
- [28] J. Gao, Y. Shi, Y. Cai, Research on the application of UAV oblique photogrammetry to LiLong housing: taking Meilan lane as an example, Int. Arch. Photogramm. Remote. Sens. Spat. Inf. Sci. 48 (2023) 629–635, https://doi.org/ 10.5194/isprs-archives-XLVIII-M-2-2023-629-2023.
- [29] G. Grenzdörffer, M. Naumann, F. Niemeyer, A. Frank, Symbiosis of UAS photogrammetry and TLS for surveying and 3D modeling of cultural heritage monuments-a case study about the cathedral of St, in: Nicholas in the City of

Greifswald, International Conference on Unmanned Aerial Vehicles in Geomatics Vol. 40, 2015, pp. 91–96, https://doi.org/10.5194/isprsarchives-XL-1-W4-33-2015. Toronto, Canada.

- [30] Y. Ham, K.K. Han, J.J. Lin, M. Golparvar-Fard, Visual monitoring of civil infrastructure systems via camera-equipped unmanned aerial vehicles (UAVs): a review of related works, Visualization in Engineering 4 (1) (2016) 1–8.
- [31] Y. Han, D. Feng, W. Wu, X. Yu, G. Wu, J. Liu, Geometric shape measurement and its application in bridge construction based on UAV and terrestrial laser scanner, Autom. Constr. 151 (2023) 104880, https://doi.org/10.1016/j. autcon.2023.104880.
- [32] Y. Han, S. Zhou, P. Xia, Q. Zhao, Research on fine 3D modeling technology of tall buildings based on UAV Photogrammetry, in: 2022 3rd International Conference on Geology, Mapping and Remote Sensing (ICGMRS), IEEE, 2022, pp. 349–353, https://doi.org/10.1109/ICGMRS55602.2022.9849357.
- [33] R. Huang, Y. Xu, L. Hoegner, U. Stilla, Semantics-aided 3D change detection on construction sites using UAV-based photogrammetric point clouds, Autom. Constr. 134 (2022) 104057, https://doi.org/10.1016/j.autcon.2021.104057.
- [34] A. Irschara, V. Kaufmann, M. Klopschitz, H. Bischof, F. Leberl, Towards fully automatic photogrammetric reconstruction using digital images taken from UAVs, in: ISPRS Technical Commission VII Symposium, 100 Years ISPRS, Advancing Remote Sensing Science Vol. XXXVIII, 2010, pp. 65–70. Part 7A, Vienna, Austria, https://www.isprs.org/proceedings/xxxviii/part7/a/proceed dings.partAweb.pdf.
- [35] Jbknowledge, in: The 4th Annual Construction Technology Report (2015), JBKnowledge, Inc, Bryan, TX, 2015. Accessed date: July 2022,, https://contechre port.com/.
- [36] Jbknowledge, in: The 9th Annual Construction Technology Report (2020), JBKnowledge, Inc, Bryan, TX, 2020. Accessed date: July 2022,, https://contechre port.com/.
- [37] G.Y. Jeong, T.N. Nguyen, D.K. Tran, T.B.H. Hoang, Applying unmanned aerial vehicle photogrammetry for measuring dimension of structural elements in traditional timber building, Measurement 153 (2020) 107386, https://doi.org/ 10.1016/j.measurement.2019.107386.
- [38] N. Kayhani, B. McCabe, A. Abdelaal, A. Heins, A.P. Schoellig, Tag-based indoor localization of uavs in construction environments: opportunities and challenges in practice, in: Construction Research Congress 2020, 2020, pp. 226–235. American Society of Civil Engineers Reston, VA, https://ascelibrary.org/doi/abs/10.1061/ 9780784482865.025.
- [39] J. Kim, D. Chung, Y. Kim, H. Kim, Deep learning-based 3D reconstruction of scaffolds using a robot dog, Autom. Constr. 134 (2022) 104092, https://doi.org/ 10.1016/j.autcon.2021.104092.
- [40] P. Kim, J. Park, Y.K. Cho, J. Kang, UAV-assisted autonomous mobile robot navigation for as-is 3D data collection and registration in cluttered environments, Autom. Constr. 106 (2019) 102918, https://doi.org/10.1016/j. autcon.2019.102918.
- [41] P. Klapa, Integration of terrestrial laser scanning and UAV-based photogrammetry for heritage Building Information Modeling, Geomatics Landmanag. Lands. (2023), https://doi.org/10.15576/GLL/2023.1.23.
- [42] L. Klein, N. Li, B. Becerik-Gerber, Imaged-based verification of as-built documentation of operational buildings, Autom. Constr. 21 (2012) 161–171, https://doi.org/10.1016/j.autcon.2011.05.023.
- [43] P. Ko, S. Prieto, B.G. de Soto, ABECIS: an automated building exterior crack inspection system using UAVs, open-source deep learning and photogrammetry, ISARC, in: Proceedings of the International Symposium on Automation and Robotics in Construction 38, IAARC Publications, 2021, pp. 637–644, https://doi. org/10.22260/ISARC2021/0086.
- [44] T. Koch, M. Körner, F. Fraundorfer, Automatic and semantically-aware 3D UAV flight planning for image-based 3D reconstruction, Remote Sens. 11 (13) (2019) 1550, https://doi.org/10.3390/rs11131550.
- [45] O. Küng, C. Strecha, P. Fua, D. Gurdan, M. Achtelik, K.-M. Doth, J. Stumpf, Simplified building models extraction from ultra-light UAV imagery, ISPRS Zurich 2011 Workshop XXXVIII-1/C22 (2011) 217–222, https://doi.org/ 10.5194/isprsarchives-XXXVIII-1-C22-217-2011.
- [46] S. Lagüela, L. Díaz, D. Roca, H. Lorenzo, Aerial thermography from low-cost UAV for the generation of thermographic digital terrain models, Opto-Electron. Rev. 23 (1) (2015) 78–84, https://doi.org/10.1515/oere-2015-0006.
- [47] N.M. Levine, B.F. Spencer Jr., Post-earthquake building evaluation using UAVs: a BIM-based digital twin framework, Sensors 22 (3) (2022) 873, https://doi.org/ 10.3390/s22030873.
- [48] F. Li, S. Zlatanova, M. Koopman, X. Bai, A. Diakité, Universal path planning for an indoor drone, Autom. Constr. 95 (2018) 275–283, https://doi.org/10.1016/j. autcon.2018.07.025.
- [49] Y. Li, C. Liu, Applications of multirotor drone technologies in construction management, Int. J. Constr. Manag. 19 (5) (2019) 401–412, https://doi.org/ 10.1080/15623599.2018.1452101.
- [50] Y. Liang, X. Fan, Y. Yang, D. Li, T. Cui, Oblique view selection for efficient and accurate building reconstruction in rural areas using large-scale UAV images, Drones 6 (7) (2022) 175, https://doi.org/10.3390/drones6070175.

- [51] Y. Liu, Y. Lin, J.K. Yeoh, D.K. Chua, L.W. Wong, M.H. Ang, W. Lee, M.Y. Chew, Framework for Automated UAV-Based Inspection of External Building Façades, Automating Cities, Springer, Singapore, 2021, pp. 173–194. Accessed date: July 2022, https://doi.org/10.1007/978-981-15-8670-5_7.
- [52] L. López-Fernández, S. Lagüela, J. Fernández, D. González-Aguilera, Automatic evaluation of photovoltaic power stations from high-density RGB-T 3D point clouds, Remote Sens. 9 (6) (2017) 631, https://doi.org/10.3390/rs9060631.
- [53] T. Luhmann, M. Chizhova, D. Gorkovchuk, Fusion of UAV and terrestrial photogrammetry with laser scanning for 3D reconstruction of historic churches in Georgia, Drones 4 (3) (2020) 53, https://doi.org/10.3390/drones4030053.
- [54] T. Luhmann, S. Robson, S. Kyle, J. Boehm, Close-Range Photogrammetry and 3D Imaging, Walter de Gruyter GmbH & Co KG, 2019. Accessed date: July 2022, https://www.degruyter.com/document/doi/10.1515/9783110302783.fm/pdf.
- [55] A. Marin, B. Wellman, Social network analysis: an introduction, in: The SAGE Handbook of Social Network Analysis, Sage Publications, 2011, pp. 11–25. Accessed date: July 2022, https://doi.org/10.4135/9781446294413. ISBN 1847873952.
- [56] P. Martínez-Carricondo, F. Carvajal-Ramírez, L. Yero-Paneque, F. Agüera-Vega, Combination of nadiral and oblique UAV photogrammetry and HBIM for the virtual reconstruction of cultural heritage. Case study of Cortijo del Fraile in Níjar, Almería (Spain), Build. Res. Inf. 48 (2) (2020) 140–159, https://doi.org/ 10.1080/09613218.2019.1626213.
- [57] B. McCabe, H. Hamledari, A. Shahi, P. Zangeneh, E.R. Azar, Roles, benefits, and challenges of using UAVs for indoor smart construction applications, in: ASCE International Workshop on Computing in Civil Engineering 2017, Seattle, Washington, 2017, pp. 349–357. https://ascelibrary.org/doi/abs/10.1061/ 9780784480830.043.
- [58] D. Moher, A. Liberati, J. Tetzlaff, D.G. Altman, P. Group, Preferred reporting items for systematic reviews and meta-analyses: the PRISMA statement, PLoS Med. 6 (7) (2009) 336–341, https://doi.org/10.1016/j.ijsu.2010.02.007.
- [59] A.A. Molina, Y. Huang, Y. Jiang, A review of unmanned aerial vehicle applications in construction management: 2016–2021, Standards 3 (2) (2023) 95–109, https://doi.org/10.3390/standards3020009.
- [60] I. Mosly, Applications and issues of unmanned aerial systems in the construction industry, Int. J. Const. Eng. Manag. 21 (6(6)) (2017) 235–239, https://doi.org/ 10.5923/j.ijcem.20170606.02.
- [61] A. Murtiyoso, M. Koehl, P. Grussenmeyer, T. Freville, Acquisition and processing protocols for UAV images: 3D modeling of historical buildings using photogrammetry, in: 26th International CIPA Symposium IV-2/W2, 2017, pp. 163–170, https://doi.org/10.5194/isprs-annals-IV-2-W2-163-2017. Ottawa, Canada.
- [62] K.K. Mwangangi, P. Mc'Okeyo, S. Oude Elberink, F. Nex, Exploring the potentials of UAV photogrammetric point clouds in facade detection and 3D reconstruction of buildings, in: XXIV ISPRS Congress, 2022 edition Vol. 43, 2022, pp. 433–440, https://doi.org/10.5194/isprs-archives-XLIII-B2-2022-433-2022. Nice, France.
- [63] R. Nagasawa, E. Mas, L. Moya, S. Koshimura, Model-based analysis of multi-UAV path planning for surveying postdisaster building damage, Sci. Rep. 11 (1) (2021) 18588, https://doi.org/10.1038/s41598-021-97804-4.
- [64] F. Nex, F. Remondino, UAV for 3D mapping applications: a review, Appl. Geomatics 6 (1) (2014) 1–15, https://doi.org/10.1007/s12518-013-0120-x.
- [65] J.M. Nwaogu, Y. Yang, A.P. Chan, H.-I. Chi, Application of drones in the architecture, engineering, and construction (AEC) industry, Autom. Constr. 150 (2023), https://doi.org/10.1016/j.autcon.2023.104827, pp. 104827.
- [66] C. Ordóñez, J. Martínez, P. Arias, J. Armesto, Measuring building façades with a low-cost close-range photogrammetry system, Autom. Constr. 19 (6) (2010) 742–749, https://doi.org/10.1016/j.autcon.2010.03.002.
- [67] O. Özyeşil, V. Voroninski, R. Basri, A. Singer, A survey of structure from motion*, Acta Numerica 26 (2017) 305–364, https://doi.org/10.1017/ S096249291700006X.
- [68] J. Park, P. Kim, Y.K. Cho, J. Kang, Framework for automated registration of UAV and UGV point clouds using local features in images, Autom. Constr. 98 (2019) 175–182, https://doi.org/10.1016/j.autcon.2018.11.024.
- [69] J.-C. Perez-Sanchez, R.-T. Mora-Garcia, V.R. Perez-Sanchez, M.-F. Cespedes-Lopez, The use of laser scanning and aerial photogrammetry as tools for the analysis of double-shell domes in the province of Alicante (Spain), in: Building Engineering Facing the Challenges of the 21st Century: Holistic Study from the Perspectives of Materials, Construction, Energy and Sustainability, Springer, 2023, pp. 677–695. Accessed date: Nov. 2023, 978-981-99-2713-5.
- [70] S. Pu, G. Vosselman, Knowledge based reconstruction of building models from terrestrial laser scanning data, ISPRS J. Photogramm. Remote Sens. 64 (6) (2009) 575–584, https://doi.org/10.1016/j.isprsjprs.2009.04.001.
- [71] L. Rabiu, A. Ahmad, Unmanned aerial vehicle photogrammetric products accuracy assessment: a review, Int. Arch. Photogramm. Remote. Sens. Spat. Inf. Sci. 48 (2023) 279–288, https://doi.org/10.5194/isprs-archives-XLVIII-4-W6-2022-279-2023.
- [72] T.S.N. Rachmawati, S. Kim, Unmanned Aerial Vehicles (UAV) integration with digital technologies toward construction 4.0: a systematic literature review, Sustainability 14 (9) (2022) 5708, https://doi.org/10.3390/su14095708.

- [73] T. Rakha, A. Gorodetsky, Review of Unmanned Aerial System (UAS) applications in the built environment: towards automated building inspection procedures using drones, Autom. Constr. 93 (2018) 252–264, https://doi.org/10.1016/j. autcon.2018.05.002.
- [74] A.F. Razali, M.F.M. Ariff, Z. Majid, A hybrid point cloud reality capture from terrestrial laser scanning and Uav-photogrammetry, Int. Arch. Photogramm. Remote. Sens. Spat. Inf. Sci. 46 (2022) 459–463, https://doi.org/10.5194/isprsarchives-XLVI-2-WI-2022-459-2022.
- [75] D. Roca, S. Lagüela, L. Díaz-Vilariño, J. Armesto, P. Arias, Low-cost aerial unit for outdoor inspection of building façades, Autom. Constr. 36 (2013) 128–135, https://doi.org/10.1016/j.autcon.2013.08.020.
- [76] V.A. Rohani, O.S. Hock, On social network web sites: definition, features, architectures and analysis tools, journal of, Comput. Eng. 1 (2009) 3–11. https ://jacr.sari.iau.ir/article_631400.html.
- [77] M. Saifizi, N. Syahirah, W.A. Mustafa, H.A. Rahim, M.W. Nasrudin, Using unmanned aerial vehicle in 3D modelling of UniCITI campus to estimate building size, J. Phys. Conf. Ser. 1962 (2021), https://doi.org/10.1088/1742-6596/1962/ 1/012057. IOP Publishing. 012057.
- [78] N.H. Sani, K.N. Tahar, G.R. Maharjan, J.C. Matos, M. Muhammad, 3D reconstruction of building model using UAV point clouds, Int. Arch. Photogramm. Remote. Sens. Spat. Inf. Sci. 43 (2022) 455–460, https://doi.org/10.5194/isprsarchives-XLIII-B2-2022-455-2022.
- [79] J. Shan, H. Zhu, R. Yu, Feasibility of accurate point cloud model reconstruction for earthquake-damaged structures using UAV-based photogrammetry, Struct. Control. Health Monit. 2023 (2023), https://doi.org/10.1155/2023/7743762.
- [80] M.-F.F. Siu, M. Lu, S. AbouRizk, Combining photogrammetry and robotic total stations to obtain dimensional measurements of temporary facilities in construction field, Visualization in Engineering 1 (1) (2013) 1–15, https://doi. org/10.1186/2213-7459-1-4.
- [81] J. Sun, B. Peng, C.C. Wang, K. Chen, B. Zhong, J. Wu, Building displacement measurement and analysis based on UAV images, Autom. Constr. 140 (2022) 104367, https://doi.org/10.1016/j.autcon.2022.104367.
- [82] Z. Sun, Y. Zhang, Using drones and 3D modeling to survey Tibetan architectural heritage: a case study with the multi-door stupa, Sustainability 10 (7) (2018) 2259, https://doi.org/10.3390/su10072259.
- [83] Y. Tan, G. Li, R. Cai, J. Ma, M. Wang, Mapping and modelling defect data from UAV captured images to BIM for building external wall inspection, Autom. Constr. 139 (2022) 104284, https://doi.org/10.1016/j.autcon.2022.104284.
- [84] Y. Tan, S. Li, H. Liu, P. Chen, Z. Zhou, Automatic inspection data collection of building surface based on BIM and UAV, Autom. Constr. 131 (2021) 103881, https://doi.org/10.1016/j.autcon.2021.103881.
- [85] Y. Tan, W. Yi, P. Chen, Y. Zou, An adaptive crack inspection method for building surface based on BIM, UAV and edge computing, Autom. Constr. 157 (2024) 105161, https://doi.org/10.1016/j.autcon.2023.105161.
- [86] M.C. Tatum, J. Liu, Unmanned aircraft system applications in construction, Proc. Eng. 196 (2017) 167–175, https://doi.org/10.1016/j.proeng.2017.07.187.
- [87] Q.K. Tran, C.H. Quan, N.K.N. Luu, K.H. Ngo, Mini-drone photogrammetry for 3D modeling of city building: a case study at Ho Chi Minh University of Technology, in: 2022 9th NAFOSTED Conference on Information and Computer Science (NICS), IEEE, 2022, pp. 351–356, https://doi.org/10.1109/ NICC556915.2022.10013448.
- [88] A. Tscharf, M. Rumpler, F. Fraundorfer, G. Mayer, H. Bischof, On the use of UAVs in mining and archaeology-geo-accurate 3d reconstructions using various platforms and terrestrial views, in: International Conference on Unmanned Aerial Vehicles in Geomatics vol. 2, 2015, pp. 15–22, https://doi.org/10.5194/ isprsannals-II-1-W1-15-2015. Toronto, Canada.
- [89] S. Tuttas, U. Stilla, A. Braun, A. Borrmann, Validation of BIM components by photogrammetric point clouds for construction site monitoring, in: PIA15+ HRIGI15 – Joint ISPRS Conference 2015 Vol. 2, 2015, pp. 231–237, https://doi. org/10.5194/isprsannals-II-3-W4-231-2015. Munich, Germany.
- [90] P. Tysiac, A. Sieńska, M. Tarnowska, P. Kedziorski, M. Jagoda, Combination of terrestrial laser scanning and UAV photogrammetry for 3D modelling and degradation assessment of heritage building based on a lighting analysis: case study—St. Adalbert Church in Gdansk, Poland, Heritage Sci. 11 (1) (2023) 1–14, https://doi.org/10.1186/s40494-023-00897-5.
- [91] M. Uysal, A. Toprak, N. Polat, Photo realistic 3D modeling with UAV: Gedik ahmet pasha mosque in afyonkarahisar, Int. Arch. Photogramm. Remote Sens. Spat. Inform. Sci. 5 (2013) W2, https://doi.org/10.5194/isprsarchives-XL-5-W2-659-2013.
- [92] G. Vacca, A. Dessì, A. Sacco, The use of nadir and oblique UAV images for building knowledge, ISPRS Int. J. Geo Inf. 6 (12) (2017) 393, https://doi.org/ 10.3390/ijgi6120393.
- [93] A. Vetrivel, M. Gerke, N. Kerle, G. Vosselman, Segmentation of UAV-based images incorporating 3D point cloud information, Int. Arch. Photogramm. Remote. Sens. Spat. Inf. Sci. (2015), https://doi.org/10.5194/isprsarchives-XL-3-W2-261-2015.
- [94] M. Videras Rodríguez, S.G. Melgar, A.S. Cordero, J.M.A. Márquez, A critical review of unmanned aerial vehicles (Uavs) use in architecture and urbanism: Scientometric and bibliometric analysis, Appl. Sci. 11 (21) (2021) 9966, https:// doi.org/10.3390/app11219966.

T. He et al.

- [95] C.-H. Wang, J.P. Mills, P.D. Gosling, B. Bridgens, R.J. Grisdale, Monitoring the testing, construction and as-built condition of membrane structures by close range photogrammetry, in: Proceedings of the ISPRS Commission V Mid-Term Symposium 'Close Range Image Measurement Techniques' 38, 2010, pp. 592–596. UK, https://www.isprs.org/proceedings/xxxviii/part5/papers/215. pdf.
- [96] J. Wang, T. Ueda, A review study on unmanned aerial vehicle and mobile robot technologies on damage inspection of reinforced concrete structures, Struct. Concr. 24 (1) (2023) 536–562, https://doi.org/10.1002/suco.202200846.
- [97] C. Xiong, Q. Li, X. Lu, Automated regional seismic damage assessment of buildings using an unmanned aerial vehicle and a convolutional neural network, Autom. Constr. 109 (2020) 102994, https://doi.org/10.1016/j. autcon.2019.102994.
- [98] F. Yamazaki, T. Matsuda, S. Denda, W. Liu, Construction of 3D models of buildings damaged by earthquakes using UAV aerial images, in: Proceedings of the Tenth Pacific Conference on Earthquake Engineering Vol. 204, 2015. Sydney, Australia, https://aees.org.au/wp-content/uploads/2015/12/Paper_204.pdf.
- [99] D.D. York, A.J. Al-Bayati, Z.Y. Al-Shabbani, Potential Applications of UAV within the Construction Industry and the Challenges Limiting Implementation, in: Construction Research Congress 2020: Project Management and Controls, Materials, and Contracts, American Society of Civil Engineers Reston, VA, 2020, pp. 31–39. https://ascelibrary.org/doi/abs/10.1061/9780784482889.004.
- [100] Z. Zhang, Review of 3D reconstruction technology of UAV aerial image, J. Phys. Conf. Ser. 1865 (2021), https://doi.org/10.1088/1742-6596/1865/4/042063. IOP Publishing. pp. 042063.
- [101] X. Zheng, F. Wang, Z. Li, A multi-UAV cooperative route planning methodology for 3D fine-resolution building model reconstruction, ISPRS J. Photogramm. Remote Sens. 146 (2018) 483–494, https://doi.org/10.1016/j. isprsjprs.2018.11.004.
- [102] S. Zhou, M. Gheisari, Unmanned aerial system applications in construction: a systematic review, Constr. Innov. (2018), https://doi.org/10.1108/CI-02-2018-0010.